Author's response to reviews

Title: ADHD in girls and boys - gender differences in co-existing symptoms and executive function measures

Authors:

Erik W Skogli (erik.skogli@sykehuset-innlandet.no)
Martin H Teicher (Martin_teicher@hms.harvard.edu)
Per N Andersen (Per.normann.andersen@sykehuset-innlandet.no)
Kjell Tore Hovik (Kjell.tore.hovik@sykehuset-innlandet.no)
Merete Øie (Merete.Oie@sykehuset-innlandet.no)

Version: 2 Date: 9 November 2012

Author's response to reviews:

On page 8 in the Methods section, we have specified that written informed consent was obtained from all participants above 12 years and all parents/caregivers. All children under the age of 12 years provided oral consent to participate.